Author’s response to reviews

Title: The short-term effect on alliance and satisfaction of using patient feedback scales in mental health out-patient treatment. A randomised controlled trial.

Authors:

Marit B Rise (marit.b.rise@ntnu.no)
Lasse Eriksen (lasse.eriksen@stolav.no)
Hilde Grimstad (hilde.grimstad@ntnu.no)
Aslak Steinsbekk (aslak.steinsbekk@ntnu.no)

Version: 3 Date: 10 September 2012

Author’s response to reviews: see over
Editor,
BMC Health Services Research

Changes to manuscript:
The short-term effect on alliance and satisfaction of using patient feedback scales in mental health out-patient treatment. A randomised controlled trial.

We are grateful for the acceptance of our manuscript. We have changed the manuscript according to your comments:

Editor’s comment:
1) Kindly provide the specific name of the ethics committee which approved the study in your manuscript.

We have re-checked the specific name of the ethics committee. In the methods section this is now stated:

“The Regional Committee for Medical and Health Research Ethics, Central Norway, and the hospital’s management approved of the study.”

Thank you for the positive response on our manuscript, and the possibility to make these changes.

Yours sincerely
Marit By Rise /s/